Author's response to reviews

Title: Utility of qualitative C-reactive protein assay and White blood cells counts in the diagnosis of neonatal septicaemia at Bugando Medical Centre, Tanzania

Authors:
Flora Chacha (chachaf733@gmail.com)
Mariam M Mirambo (mmmirambo@gmail.com)
Martha M Mushi (marthamushi@yahoo.com)
Neema Kayange (n_eema@yahoo.com)
Antke Zuechner (antke.zuechner@web.de)
Benson R Kidenya (benkidenya@yahoo.com)
Stephen E. Mshana (mshana72@yahoo.com)

Version: 4 Date: 25 September 2014

Author's response to reviews: see over
Dear EIC

BMC Pediatrics

1. Copyediting:
   The manuscript has extensive reviewed by native English speaker

2. Title Page: Please include the email addresses of all authors on the title page. Please ensure that it is identical to the one entered on the submission system.
   Missing email of Martha Mushi has been provided

3. Consent: Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.
   The statement has provided as seen in page 7

Prof Stephen Mshana
On behalf of all authors